

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL105046

Key Points:

- Autocorrelation of clean Seismic Experiment for Interior Structure waveforms recovers the prominent signals reflected from deep Mars
- The autocorrelation tests on raw, clean and glitch-only data suggest the signals identified are from deep Mars rather than instrumental noise
- A velocity transition model of the Martian core-mantle boundary better fits the observed PcP phase than a sharp velocity drop

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

S. Deng, sd62@rice.edu; szhdeng@gmail.com

Citation:

Deng, S., & Levander, A. (2023). Seismic autocorrelation analysis of deep Mars. *Geophysical Research Letters*, 50, e2023GL105046. https://doi. org/10.1029/2023GL105046

Received 16 JUN 2023 Accepted 27 NOV 2023

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Seismic Autocorrelation Analysis of Deep Mars

Sizhuang Deng^{1,2} 💿 and Alan Levander¹ 💿

¹Department of Earth, Environmental and Planetary Sciences, Rice University, Houston, TX, USA, ²Now at Chevron Technical Center, a Division of Chevron U.S.A. Inc, Houston, TX, USA

Abstract The InSight mission deployed one seismic station on Mars, providing a chance to apply single-station-based autocorrelation analysis to investigate Martian subsurface structures. However, recent analysis indicated the low-frequency autocorrelation signals may originate from quasi-periodic high-amplitude instrumental "glitches" rather than the reflection response of deep Mars. In this study, we detected and removed these high-amplitude glitches in raw seismic data and employed autocorrelation on the clean vertical component waveforms filtered between 0.05 and 0.1 Hz. We observed signals at the expected times for the olivine-wadsleyite transition and core-mantle boundary (CMB) as estimated by other methods. This result suggests that the low-frequency autocorrelation signals are the reflection response from the olivine-wadsleyite transition in the mantle and the Martian CMB region, rather than a noise phenomena. A grid search method to fit the observed PcP waveform was used to identify a layer intermediate in velocity between the Martian mantle and core at the Martian CMB.

Plain Language Summary The InSight mission deployed one seismic station on Mars at the end of 2018 to measure the interior structure of Mars. A recent study autocorrelated the InSight ambient noise data to retrieve reflection signals from subsurface discontinuities. Two seismic signals that reflected from deep within Mars, the olivine-wadsleyite transition, a pressure-temperature dependent mineral phase change in the mantle, and the core-mantle boundary (CMB), are observed. However, some analyses suggested that the arrival times of these signals coincide with the recurrence time of high-amplitude glitches within the raw seismic data, leading to an incorrect interpretation of the autocorrelation function (ACF). To resolve this issue, we detected and removed the high-amplitude glitches before further processing the ambient noise data. The autocorrelation analysis of clean continuous vertical-component waveforms recovers signals at times corresponding to the olivine-wadsleyite transition and CMB, whereas ACF of glitch-only waveforms does not. This suggests that signals on the low-frequency ACF are from the seismic discontinuities at olivine-wadsleyite transition and CMB rather than being noise artifacts. We further identified velocity layering near Martian CMB by matching the complexity of the PcP signal observed in the ACF of clean data set with synthetic seismograms.

1. Introduction

The spacecraft of the InSight (Interior Exploration Using Seismic Investigations, Geodesy and Heat Transport) mission landed on Mars at the end of 2018 (Banerdt et al., 2020). In early 2019, it deployed the only seismic station, SEIS (Seismic Experiment for Interior Structure), to record the seismic vibrations on Mars (Lognonné et al., 2019). The InSight seismometer successfully detected ~1,300 Marsquake events before its termination in December 2022 (Ceylan et al., 2023), providing indications of Martian tectonic activity (Ceylan et al., 2022; Drilleau et al., 2022; Giardini et al., 2020; Horleston et al., 2022; Jacob et al., 2022; Sun & Tkalčić, 2022). These seismic data have been used to determine the interior structures of the red planet at different scales using surface wave analysis (Beghein et al., 2022; Deng & Levander, 2022; Kim et al., 2022; Li et al., 2023; Xu et al., 2023), autocorrelation functions (ACFs) (Compaire et al., 2021; Deng & Levander, 2020, 2022; Qin et al., 2023; Schimmel et al., 2021), receiver functions (Knapmeyer-Endrun et al., 2021; Lognonné et al., 2020), anisotropy analysis (Li et al., 2022), attenuation modeling (Karakostas et al., 2021) and geophysical inversion (Khan et al., 2021; Stähler et al., 2021). Lognonné, Banerdt, et al. (2023) provide a review of InSight seismic research.

Here we use autocorrelation analysis of the SEIS data to identify reflections from the Martian mantle and core-mantle boundary (CMB) region. Claerbout (1968) showed that the reflectivity response for a coincident surface source and receiver in an acoustic medium can be reconstructed by autocorrelation of the normal-incident plane wave response due to a source incident from below recorded by a surface receiver. This was later extended to elastic media and non-normal incidence cases (Frasier, 1970). Autocorrelation analysis of single-station seismic



recordings (e.g., ambient noise, coda waves) has been widely employed to extract body-wave reflection phases to reveal the subsurface structures of the Earth (Delph et al., 2019; Gorbatov et al., 2013; Kennett, 2015; Oren & Nowack, 2017; Phạm & Tkalčić, 2018, 2021; Qin et al., 2020; She et al., 2022; Tibuleac & von Seggern, 2012; Zhou & Zhang, 2021) and Moon (Nishitsuji et al., 2016, 2020). Recent autocorrelation analysis on Mars detected a strong P-wave reflection phase at 10.5–11.5 s two-way traveltime (Compaire et al., 2021; Deng & Levander, 2020; Schimmel et al., 2021). This 10.5–11.5 s signal can be interpreted as the reflection from either a mid-crust discontinuity or the Martian Moho at approximately 35 km (Deng & Levander, 2020). To examine deeper seismic structure of Mars, Deng and Levander (2020), hereafter referred to as DL2020, filtered the stacked autocorrelations in the low-frequency 0.05 and 0.1 Hz band and interpreted signals at \sim 280 and \sim 375 s as P-wave reflections from the Martian olivine-wadsleyite transition and CMB, respectively. Depth conversion using different reference velocity models determined the Martian core radius in the range of 1,790-1,870 km (Deng & Levander, 2020), consistent with the core radius estimation from other studies (Khan et al., 2018; Rivoldini et al., 2011; Yoder et al., 2003). However, Lognonné et al. (2020) identified high-amplitude seismic glitches within raw SEIS data which probably originate from stress relaxation within the InSight seismometer. Scholz et al. (2020) subsequently analyzed the glitch properties. These glitches may contaminate the final results of seismic analysis if not removed (Compaire et al., 2021). Barkaoui et al. (2021) analyzed the quasi-periodic recurrence time of these seismic glitches and suggest that it coincides with the arrival times of identified in low-frequency autocorrelations in DL2020. Kim et al. (2021) also did autocorrelation tests that the signals interpreted as olivine-wadsleyite transition and CMB in DL2020 may be contaminated by the quasi-periodic high-amplitude glitches. As more seismic data on Mars have been collected since DL2020, further analysis of deep Martian structure is warranted.

In this work, we designed a test to resolve whether the low-frequency autocorrelation signals in DL2020 are real seismic reflection signals or result from instrumental noise. The high-amplitude glitches in raw SEIS data were first detected to generate three data sets for further analysis: (a) the raw continuous data with glitches; (b) the data with only detected glitches; (c) the clean continuous data after the application of glitch removal algorithms, hereafter referred to as the "cleaned" or "deglitched" data set (Scholz et al., 2020). We autocorrelated the ambient noise recorded on the vertical-components of these three data sets, raw, glitch only, and cleaned, to demonstrate that the low-frequency autocorrelation signals reported by DL2020 are the reflection response from deep Mars rather than the recurrence time of quasi-periodic glitches. Using the velocity model of Stähler et al. (2021) we calculated synthetic PcP seismograms to model the complexity of the PcP phase observed in the cleaned autocorrelation. A layer of intermediate velocity between the Martian mantle and core at the Martian CMB was identified using a grid search to minimize the misfit between the synthetic and observed PcP phases.

2. Data and Methods

The SEIS data were collected with a 10 and 20 Hz sampling rate in two different periods. Prior to preprocessing, the open-source package SEISglitch (Scholz et al., 2020) was applied to extract and remove the high-amplitude glitches from the raw continuous SEIS waveforms to produce cleaned and glitch-only data. Figure 1a–1c provide a 1-day comparison of original and cleaned continuous U-, V- and W-component data. The glitches were mostly detected at night time during periods of relatively low ground temperature (Figure 1d) and windspeed (Figure 1e) because of the lower wind- and pressure-driven noise level during the nighttime (Scholz et al., 2020). Figures 1g and 1h show the spectrograms of original and cleaned W-component waveforms in a 3-hr window (Figure 1f). The glitches are recorded as high-amplitude vertical peaks in spectrogram (Figure 1g). The glitch energy is effectively reduced by the glitch removal algorithm within SEISglitch package (Figure 1h). Figures S1 and S2 in Supporting Information S1 describe parameter selection test for glitch detection and removal, which suggest that parameter values used in SEISglitch have modest impact on the final autocorrelation results. The preprocessing procedures included the instrument response removal from the raw, clean and glitch-only continuous SEIS data with a broad, 0.01–3.5 Hz, bandpass filter. Data sampled at 20 Hz were decimated to 10 Hz. The continuous waveforms were cut into 4-hr-long segments and rotated from U-V-W into orthogonal Z-N-E channels (Compaire et al., 2021). We removed the mean and trend and applied a taper to each 4-hr window.

We used a workflow similar to Bensen et al. (2007) to calculate ACFs for the 3 data sets. Temporal balance and spectral whitening were employed to improve the interpretability of the stacked ambient noise ACFs (Bensen et al., 2007; Oren & Nowack, 2017). Each individual 4-hr trace was then filtered from 0.05 to 0.1 Hz. We computed and normalized the vertical component autocorrelograms by the amplitude at zero lag time for each 4-hr trace. We applied both linear (Linearly stacked (LS)) and phase-weighted (PWS) stacking (Schimmel & Paulssen, 1997) to produce stacked

10.1029/2023GL105046



Figure 1. (a) Original (blue) and Cleaned (orange) U-component waveforms on 3 January 2020. (b) Same as (a) but for V-component. (c) Same as (a) but for W-component. (d) Corresponding ground temperature data and (e) windspeed data on. (f) Original (blue) and Cleaned (orange) W-component waveforms in a 3-hr window (3 January 2020 03:00:00 UTC to 3 January 2020 06:00:00 UTC) shown as the red box in (c). (g) The spectrogram of original W-component waveform in (f). The black arrows identify glitch examples, which are characterized as broadband (i.e., vertical) features within the spectrogram. (h) The spectrogram of cleaned W-component waveform in (f), where the reduction of glitch energy is clear in comparison to the spectrogram in (g).

autocorrelograms. The power used for PWS, which defines the significance of coherency measure (Schimmel & Paulssen, 1997), was empirically selected as 2 (Korenaga, 2014; Niu & Chen, 2008; Wookey & Helffrich, 2008).

3. Results and Discussion

3.1. Autocorrelation Results

Figure 2 compares the LS and PWS stacked autocorrelation of raw, cleaned, glitch-only and 80% glitch-only data sets. The 80% glitch-only data set consists of ~80% of largest glitches within glitch-only data set. We made this

DENG AND LEVANDER





Figure 2. (a) Linearly stacked vertical-component autocorrelation filtered between 0.05 and 0.1 Hz using raw (black), clean (red), glitch-only (blue) and 80% glitch-only (green) data. The gray dashed lines mark the reflection phases from olivine-wadsleyite transition and core-mantle boundary. (b) Same as (a) but for phase-weighted stacking (Schimmel & Paulssen, 1997).

comparison between glitch-only and 80% glitch to ascertain the importance of small glitches left by the glitch removal algorithm (Scholz et al., 2020). The stacked ACFs from these two data sets are almost identical for both LS (Figure 2a) and PWS (Figure 2b), which suggests the smaller amplitude glitches remaining in the clean data set will have modest impact on final ACFs.

Two signals at \sim 285 and \sim 385 s are clearly observed in the stacked ACF. The signal at \sim 285 s in the clean ACF (red curves in Figures 2a and 2b) is consistent with the traveltime of the P-wave reflection from the olivine-wadsleyite transition as calculated for a number of Martian velocity models (Khan et al., 2023; Stähler et al., 2021).

We turn our attention to the analysis of the interpreted PcP phase at ~385 s, leaving further discussion of the ~285 s signal to the Supporting Information S1 (Text S4 and Figure S5). The arrival time of this signal is consistent with the prominent phase identified as the P-wave reflection from CMB reported in DL2020. The bootstrap calculation also indicates the observation of this PcP signal is robust (Figure S4 in Supporting Information S1). The ACFs of glitch-only and 80% glitch-only data do not show a prominent response at ~385 s. We take this as indicating the reflection response at ~385 s originated from Martian CMB rather than an unfortunate correlation of glitches suggested by Barkaoui et al. (2021) and Kim et al. (2021). The noise sources to generate this deep Mars reflection signal may come from atmospheric phenomena (Nishikawa et al., 2019) or coda waves of Marsquakes (Lognonné, Schimmel, et al., 2023), which are less attenuated during wave propagation as Mars is a dry, cold and tectonically inactive planet compared with Earth (Menina et al., 2021). In the stacked raw data ACF there is a signal at ~385 s, but the reflection pulse is much more complicated than in the clean waveform because it is contaminated by the high-amplitude seismic glitches in the raw data. The PWS stacked autocorrelation of the deglitched ambient noise data (red curve in Figure 2b) will be used for the comparison with synthetic results since the observed reflection phases are clearer and simpler than the linear stack.

3.2. Comparison With Synthetic PcP Phases

We simulated the synthetic normal-incident PcP phases with Thomson-Haskell matrix method (Haskell, 1962; Thomson, 1950) for a set of velocity models (Figure 3a) from a probabilistic inversion of seismic traveltimes,





Figure 3. (a) Mean velocity model and 3 standard deviations above and below the mean model in Stähler et al. (2021). The solid lines are P-wave velocity profiles, and the dashed lines are S-wave velocity profiles. (b) Comparison between the synthetic PcP phases for the velocity models shown in (a) and the observed PcP phase from the autocorrelation of the clean ambient noise data (red curve in Figure 2b). The lag time represents the correlation time shift between the synthetic and observed PcP phases, where the positive values represent the observed PcP travels slower than synthetic PcP, and vice versa.

Love number k_2 and moment of inertia (Stähler et al., 2021). The comparison between synthetic and real PcP signals (Figure 3b) shows that the velocity model 3 standard deviations lower than the mean model of Stähler et al. (2021) best matches the timing of the observed PcP phase. This observation is consistent with the Mars orbiting surface wave results retrieved from ambient noise autocorrelation (Deng & Levander, 2022). We will use Stähler et al. (2021)'s mean-3 σ model (Figure 3a) as a reference in later analysis.

3.3. Grid Search for CMB Transition Zone Modeling

The PcP reflection waveform at ~385 s is more complicated than that from the potential olivine-wadsleyite transition at ~ 285 s (red curve in Figure 2b), therefore we model the CMB as a transition zone (TZ), with the velocity $V_{\text{TZ}} = \alpha V_{\text{mantle}} + (1 - \alpha)V_{\text{core}}$, where $0 \le \alpha \le 1$. For $\alpha = 1$, this is the mean- 3σ velocity model shown in Figure 3a which best fits the observed PcP phase among the Stähler et al. (2021) suite of models (Figure 3b). Two parameters, the TZ thickness and α value, were determined by grid search to find the best combination that fits the observed PcP waveform shown in Figure 2b. The TZ thickness is set as 40, 60, 80, 100, and 120 km and the α value ranges from 0.05 to 1 with 0.05 grid size. For each TZ thickness and α value combination, we simulated the synthetic zero-offset PcP phase by Thomson-Haskell matrix (Haskell, 1962; Thomson, 1950) and calculated the correlation coefficients between the synthetic and observed PcP phases. Figures S6 to S10 in Supporting Information S1 showed the correlation coefficients and waveform comparisons with the observed PcP phase for 40, 60, 80, 100, and 120 km thick CMB transition zone models, respectively.

Figure 4a illustrates the 40, 60, 80, 100, and 120 km thick CMB transition zone models that best match with the observed PcP phase. Figure 4b makes the comparisons between the observed PcP and the synthetics of the velocity models shown in Figure 4b. We can see the 60 km thick transition zone model (green model in Figure 4a) at the Martian CMB with $V_{TZ} = 8.05$ km/s best fits the observed PcP phase (Figure 4b). Since a 60 km thick 1-layer transition zone produces a waveform correlation coefficient of 0.984 (Figure 4b), hence increasing the model complexity of CMB transition zone is unnecessary.

3.4. Possible Cause of CMB Transition Zone

A similar ultralow-velocity zone (ULVZ) near CMB has been observed on Earth across different regions, such as Southeast Asia (Yao & Wen, 2014), Hawaii (Cottaar & Romanowicz, 2012), Mexico (Havens & Revenaugh, 2001), Africa and Eastern Atlantic (Helmberger et al., 2000). ULVZs normally are several tens of kilometers thick and may yield P-wave and S-wave velocity reduction by up to ~10% and ~30%, respectively (McNamara, 2019). In general, two theories are widely acknowledged to interpret the existence of ULVZ at Earth's lower mantle:





Figure 4. (a) The models for 40, 60, 80, 100 and 120 km thick transition zones that best match with the observed PcP phase (red curves in Figure 2b). (b) The comparison between the synthetic PcP phase of the velocity models shown in (a) and the observed PcP phase. The correlation coefficients between the synthetic and observed PcP are shown on the right side. The blue box marks the model with highest correlation coefficient among all models shown in (a) (The 60 km thick 1-layer core-mantle boundary transition zone model, with $V_{TZ} = 8.05$ km/s).

(a) partial melt occurs at the location of ULVZ other than the ambient mantle (Ohtani & Maeda, 2001; Wen & Helmberger, 1998; Williams & Garnero, 1996); (b) core rust is formed when iron interacts with water or hydroxide minerals at the lower mantle (Hu et al., 2016; Liu et al., 2017; Mao et al., 2017). The low-velocity layer near the Martian CMB observed in our study (Figure 4) may follow the same theories as the ULVZ observed on Earth.

Two recent studies have presented several different Martian velocity models with a basal melt layer consisting of a low velocity silicate melt, 100–175 km in thickness and 4.5–5.5 km/s in P-wave velocity (Khan et al., 2023; Samuel et al., 2023), where the transition zone thickness is thicker than our estimation of 60 km and transition zone P-wave velocity is lower than our estimation of 8.05 km/s. To better constrain the structure and understand the cause of the CMB transition zone on Mars, advanced studies on high-pressure experiment, mineralogical simulation, and thermodynamic modeling are required to cross-validate with the seismic observations.

4. Conclusion

Kim et al. (2021) and Barkaoui et al. (2021) suggested that the autocorrelation signals identified as the reflection responses from olivine-wadsleyite transition and CMB (Deng & Levander, 2020) may originate from the quasi-periodic glitches rather than real seismic discontinuities. In this study, we applied the autocorrelation method on raw, clean, and glitch-only vertical-component data to investigate these claims. The prominent signal at ~385 s was extracted from the autocorrelation of raw and clean waveforms but not from glitch-only data (Figure 2). This suggests the signal at ~385 s was the reflection response from seismic boundaries within Mars rather than the recurrence time of high-amplitude glitches in raw SEIS data. We interpret the ~385 s signal as the CMB. We interpret an earlier phase at ~285 s (Figure 2) as the P-wave reflection at olivine-wadsleyite transition as in DL2020 (Text S4 in Supporting Information S1). The noise sources to produce deep Mars reflection signals may originate from atmospheric phenomena (Nishikawa et al., 2019) and/or Marsquake events (Lognonné, Schimmel, et al., 2023). The clarity of these signals likely results from the low seismic attenuation of the Martian interior (Menina et al., 2021). Timing of the PcP phase (~385 s) is best matched using Stähler et al. (2021)'s mean- 3σ model (Figure 3b), consistent with Mars orbiting surface wave R2 results (Deng & Levander, 2022). We then calculated synthetic PcP phases for a velocity transition zone at the CMB in which we varied velocity and layer thickness to model the observed complexity of the PcP waveform (Figure 4). The grid search shows

19448007, 2023, 24, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2023GL105046, Wiley Online Library on [13/12/2023]. See the Terms and Condition (https://or conditions) on Wiley Online Library for rules of use; OA articles

that a 60 km thick layer with $V_{TZ} = 8.05$ km/s (green model in Figure 4a) produces synthetic seismograms that best match the observed PcP waveform (Figure 4b). Samuel et al. (2023) and Khan et al. (2023) derived a similar structure using InSight seismic data, which is interpreted as the molten silicate layer at the base of Martian mantle. The observed CMB transition zone on Mars (Figure 4a) can potentially enhance our understanding of geological history and dynamic evolution of Mars (Zhang et al., 2023).

Data Availability Statement

The InSight seismic data (InSight Mars SEIS Data Service, 2019) used in this study are available on IRIS (Incorporated Research Institutions for Seismology) data center. The open-source Python package SEISglitch (Scholz et al., 2020) were used to detect and remove glitches within InSight seismic data. The ground temperature data on Mars were recorded by a radiometer on the Heat Flow and Physical Properties Package (HP3) (Mueller et al., 2020; Spohn et al., 2018). The windspeed data were recorded by Temperature and Wind Sensor (TWINS) (Banfield et al., 2019, 2020). The velocity models in Stähler et al. (2021) were used to generate synthetics and make comparison with the observed PcP phase. Several procedures used Obspy (Krischer et al., 2015) to process the seismic data.

Acknowledgments

This study was funded by the Department of Earth, Environmental and Planetary Sciences at Rice University. We thank Doveon Kim, Paul Davis, Mark Panning, John-Robert Scholz, Martin Schimmel and other members of the InSight mission for sharing their understandings about the autocorrelation results and the instrumental noise of SEIS data. We thank Simon Stähler for providing the velocity models in Stähler et al. (2021). We also thank Fenglin Niu at Rice University for discussions related to this work. Constructive comments from Editor Kevin Lewis, Reviewer Philippe Lognonné and an anonymous reviewer are appreciated.

References

- Banerdt, W. B., Smrekar, S. E., Banfield, D., Giardini, D., Golombek, M., Johnson, C. L., et al. (2020). Initial results from the InSight mission on Mars. Nature Geoscience, 13(3), 183–189. https://doi.org/10.1038/s41561-020-0544-y
- Banfield, D., Rodriguez-Manfredi, J. A., Russell, C. T., Rowe, K. M., Leneman, D., Lai, H. R., et al. (2019). InSight Auxiliary Payload Sensor Suite (APSS) [Dataset]. Space Science Reviews, 215(1), 4. https://doi.org/10.1007/s11214-018-0570-x
- Banfield, D., Spiga, A., Newman, C., Forget, F., Lemmon, M., Lorenz, R., et al. (2020). The atmosphere of Mars as observed by InSight [Dataset]. Nature Geoscience, 13(3), 190–198. https://doi.org/10.1038/s41561-020-0534-0
- Barkaoui, S., Lognonné, P., Kawamura, T., Stutzmann, É., Seydoux, L., de Hoop, M. V., et al. (2021). Anatomy of continuous Mars SEIS and pressure data from unsupervised learning. Bulletin of the Seismological Society of America, 111(6), 2964–2981. https://doi.org/10.1785/0120210095
- Beghein, C., Li, J., Weidner, E., Maguire, R., Wookey, J., Lekić, V., et al. (2022). Crustal anisotropy in the Martian Lowlands from surface waves. *Geophysical Research Letters*, 49(24), e2022GL101508. https://doi.org/10.1029/2022GL101508
- Bensen, G. D., Ritzwoller, M. H., Barmin, M. P., Levshin, A. L., Lin, F., Moschetti, M. P., et al. (2007). Processing seismic ambient noise data to obtain reliable broad-band surface wave dispersion measurements. *Geophysical Journal International*, 169(3), 1239–1260. https://doi. org/10.1111/j.1365-246X.2007.03374.x
- Ceylan, S., Clinton, J. F., Giardini, D., Stähler, S. C., Horleston, A., Kawamura, T., et al. (2022). The marsquake catalogue from InSight, sols 0–1011. *Physics of the Earth and Planetary Interiors*, 333, 106943. https://doi.org/10.1016/j.pepi.2022.106943
- Ceylan, S., Giardini, D., Clinton, J., Kim, D., Khan, A., Stähler, S. C., et al. (2023). Mapping the seismicity of Mars with InSight. Journal of Geophysical Research: Planets, 128(8), e2023JE007826. https://doi.org/10.1029/2023JE007826
- Claerbout, J. F. (1968). Synthesis of a layered medium from its acoustic transmission response. *Geophysics*, 33(2), 264–269. https://doi.org/10.1190/1.1439927
- Compaire, N., Margerin, L., Garcia, R. F., Pinot, B., Calvet, M., Orhand-Mainsant, G., et al. (2021). Autocorrelation of the ground vibrations recorded by the SEIS-InSight seismometer on Mars. *Journal of Geophysical Research: Planets*, 126(4), 1–20. https://doi.org/10.1029/2020JE006498
- Cottaar, S., & Romanowicz, B. (2012). An unusually large ULVZ at the base of the mantle near Hawaii. Earth and Planetary Science Letters, 355–356, 213–222. https://doi.org/10.1016/j.epsl.2012.09.005
- Delph, J. R., Levander, A., & Niu, F. (2019). Constraining crustal properties using receiver functions and the autocorrelation of earthquake-Generated body waves. Journal of Geophysical Research: Solid Earth, 124(8), 8981–8997. https://doi.org/10.1029/2019JB017929
- Deng, S., & Levander, A. (2020). Autocorrelation reflectivity of Mars. *Geophysical Research Letters*, 47(16). https://doi.org/10.1029/2020GL089630
 Deng, S., & Levander, A. (2022). Autocorrelation R² on Mars. *Geophysical Research Letters*, 49(17), e2022GL099580. https://doi.org/10.1029/2022GL099580
- Drilleau, M., Samuel, H., Garcia, R. F., Rivoldini, A., Perrin, C., Michaut, C., et al. (2022). Marsquake locations and 1-D seismic models for Mars from InSight data. *Journal of Geophysical Research: Planets*, 127(9). https://doi.org/10.1029/2021JE007067
- Frasier, C. W. (1970). Discrete time Solution of plane P-SV waves in a plane layered medium. *Geophysics*, 35(2), 197–219. https://doi.org/10.1190/1.1440085
- Giardini, D., Lognonné, P., Banerdt, W., Pike, W., Christensen, U., Ceylan, S., et al. (2020). The seismicity of Mars. *Nature Geoscience*, 13(3), 205–212. https://doi.org/10.1038/s41561-020-0539-8
- Gorbatov, A., Saygin, E., & Kennett, B. L. N. (2013). Crustal properties from seismic station autocorrelograms. *Geophysical Journal Interna*tional, 192(2), 861–870. https://doi.org/10.1093/gji/ggs064
- Haskell, N. A. (1962). Crustal reflection of plane P and SV waves. Journal of Geophysical Research, 67(12), 4751–4768. https://doi.org/10.1029/jz067i012p04751
- Havens, E., & Revenaugh, J. (2001). A broadband seismic study of the lowermost mantle beneath Mexico: Constraints on ultralow velocity zone elasticity and density. *Journal of Geophysical Research*, 106(B12), 30809–30820. https://doi.org/10.1029/2000jb000072
- Helmberger, D., Ni, S., Wen, L., & Ritsema, J. (2000). Seismic evidence for ultralow-velocity zones beneath Africa and eastern Atlantic. Journal of Geophysical Research, 105(B10), 23865–23878. https://doi.org/10.1029/2000jb900143
- Horleston, A. C., Clinton, J. F., Ceylan, S., Giardini, D., Charalambous, C., Irving, J. C. E., et al. (2022). The far Side of Mars: Two distant marsquakes detected by InSight. *The Seismic Record*, 2(2), 88–99. https://doi.org/10.1785/0320220007
- Hu, Q., Kim, D. Y., Yang, W., Yang, L., Meng, Y., Zhang, L., & Mao, H. K. (2016). FeO2 and FeOOH under deep lower-mantle conditions and Earth's oxygen-hydrogen cycles. *Nature*, 534(7606), 241–244. https://doi.org/10.1038/nature18018

are governed by

the applicable Creative Complexity

- InSight Mars SEIS Data Service. (2019). SEIS raw data, Insight Mission [Dataset]. IPGP, JPL, CNES, ETHZ, ICL, MPS, ISAE-Supaero, LPG, MFSC. https://doi.org/10.18715/SEIS.INSIGHT.XB_2016
- Jacob, A., Plasman, M., Perrin, C., Fuji, N., Lognonné, P., Xu, Z., et al. (2022). Seismic sources of InSight marsquakes and seismotectonic context of Elysium Planitia, Mars. *Tectonophysics*, 837, 229434. https://doi.org/10.1016/j.tecto.2022.229434
- Karakostas, F., Schmerr, N., Maguire, R., Huang, Q., Kim, D., Lekic, V., et al. (2021). Scattering attenuation of the Martian interior through codawave analysis. Bulletin of the Seismological Society of America, 111(6), 3035–3054. https://doi.org/10.1785/0120210253
- Kennett, B. L. N. (2015). Lithosphere-asthenosphere P-wave reflectivity across Australia. Earth and Planetary Science Letters, 431, 225–235. https://doi.org/10.1016/j.epsl.2015.09.039
- Khan, A., Ceylan, S., van Driel, M., Giardini, D., Lognonné, P., Samuel, H., et al. (2021). Upper mantle structure of Mars from InSight seismic data. Science, 373(6553), 434–438. https://doi.org/10.1126/science.abf2966
- Khan, A., Huang, D., Durán, C., Sossi, P. A., Giardini, D., & Murakami, M. (2023). Evidence for a liquid silicate layer atop the Martian core. *Nature*, 622(7984), 718–723. https://doi.org/10.1038/s41586-023-06586-4
- Khan, A., Liebske, C., Rozel, A., Rivoldini, A., Nimmo, F., Connolly, J. A. D., et al. (2018). A geophysical perspective on the bulk composition of Mars. Journal of Geophysical Research: Planets, 123(2), 575–611. https://doi.org/10.1002/2017JE005371
- Kim, D., Banerdt, W. B., Ceylan, S., Giardini, D., Lekić, V., Lognonné, P., et al. (2022). Surface waves and crustal structure on Mars. Science, 378(6618), 417–421. https://doi.org/10.1126/science.abq7157
- Kim, D., Davis, P., Lekić, V., Maguire, R., Compaire, N., Schimmel, M., et al. (2021). Potential pitfalls in the analysis and structural interpretation of Mars' seismic data from InSight. Bulletin of the Seismological Society of America, 1–21. https://doi.org/10.1785/0120210123
- Knapmeyer-Endrun, B., Panning, M. P., Bissig, F., Joshi, R., Khan, A., Kim, D., et al. (2021). Thickness and structure of the Martian crust from InSight seismic data. *Science*, 373(6553), 438–443. https://doi.org/10.1126/science.abf8966
- Korenaga, J. (2014). Teleseismic migration with dual bootstrap stack. Geophysical Journal International, 196(3), 1706–1723. https://doi.org/10.1093/gji/ggt475
- Krischer, L., Megies, T., Barsch, R., Beyreuther, M., Lecocq, T., Caudron, C., & Wassermann, J. (2015). ObsPy: A bridge for seismology into the scientific Python ecosystem [Software]. Computational Science & Discovery, 8(1), 014003. https://doi.org/10.1088/1749-4699/8/1/014003
- Li, J., Beghein, C., Lognonné, P., McLennan, S. M., Wieczorek, M. A., Panning, M. P., et al. (2023). Different Martian crustal seismic velocities across the dichotomy boundary from multi-orbiting surface waves. *Geophysical Research Letters*, 50(1), e2022GL101243. https://doi. org/10.1029/2022GL101243
- Li, J., Beghein, C., Wookey, J., Davis, P., Lognonné, P., Schimmel, M., et al. (2022). Evidence for crustal seismic anisotropy at the InSight lander site. Earth and Planetary Science Letters, 593, 117654. https://doi.org/10.1016/j.epsl.2022.117654
- Liu, J., Hu, Q., Young Kim, D., Wu, Z., Wang, W., Xiao, Y., et al. (2017). Hydrogen-bearing iron peroxide and the origin of ultralow-velocity zones. *Nature*, 551(7681), 494–497. https://doi.org/10.1038/nature24461
- Lognonné, P., Banerdt, W., Pike, W. T., Giardini, D., Christensen, U., Garcia, R. F., et al. (2020). Constraints on the shallow elastic and anelastic structure of Mars from InSight seismic data. *Nature Geoscience*, 13(3), 213–220. https://doi.org/10.1038/s41561-020-0536-y
- Lognonné, P., Banerdt, W. B., Clinton, J., Garcia, R. F., Giardini, D., Knapmeyer-Endrun, B., et al. (2023). Mars Seismology. Annual Review of Earth and Planetary Sciences, 51(1), 643–670. Annual Reviews. https://doi.org/10.1146/annurev-earth-031621-073318
- Lognonné, P., Banerdt, W. B., Giardini, D., Pike, W. T., Christensen, U., Laudet, P., et al. (2019). SEIS: Insight's seismic experiment for internal structure of Mars. Space Science Reviews, 215(1), 12. https://doi.org/10.1007/s11214-018-0574-6
- Lognonné, P., Schimmel, M., Stutzmann, E., Davis, P., Drilleau, M., Sainton, G., et al. (2023). Detection of Mars normal modes from S1222a event and seismic hum. *Geophysical Research Letters*, 50(12), e2023GL103205. https://doi.org/10.1029/2023gl103205
- Mao, H. K., Hu, Q., Yang, L., Liu, J., Kim, D. Y., Meng, Y., et al. (2017). When water meets iron at Earth's core-mantle boundary. National Science Review, 4(6), 870–878. https://doi.org/10.1093/nsr/nwx109
- McNamara, A. K. (2019). A review of large low shear velocity provinces and ultra low velocity zones. *Tectonophysics*, 760, 199–220. https://doi. org/10.1016/j.tecto.2018.04.015
- Menina, S., Margerin, L., Kawamura, T., Lognonné, P., Marti, J., Drilleau, M., et al. (2021). Energy envelope and attenuation characteristics of high-frequency (HF) and very-high-frequency (VF) Martian events. *Bulletin of the Seismological Society of America*, 111(6), 3016–3034. https://doi.org/10.1785/0120210127
- Mueller, N. T., Knollenberg, J., Grott, M., Kopp, E., Walter, I., Krause, C., et al. (2020). Calibration of the HP3 radiometer on InSight [Dataset]. Earth and Space Science, 7(5). https://doi.org/10.1029/2020EA001086
- Nishikawa, Y., Lognonné, P., Kawamura, T., Spiga, A., Stutzmann, E., Schimmel, M., et al. (2019). Mars' background free oscillations. In Space science reviews. Springer Netherlands. https://doi.org/10.1007/s11214-019-0579-9
- Nishitsuji, Y., Rowe, C. A., Wapenaar, K., & Draganov, D. (2016). Reflection imaging of the Moon's interior using deep-moonquake seismic interferometry. *Journal of Geophysical Research: Planets*, 121(4), 695–713. https://doi.org/10.1002/2015JE004975
- Nishitsuji, Y., Ruigrok, E., & Draganov, D. (2020). Azimuthal anisotropy of the megaregolith at the Apollo 14 landing site. Journal of Geophysical Research: Planets, 125(5), e2019JE006126. https://doi.org/10.1029/2019JE006126
- Niu, F., & Chen, Q. F. (2008). Seismic evidence for distinct anisotropy in the innermost inner core. *Nature Geoscience*, 1(10), 692–696. https://doi.org/10.1038/ngeo314
- Ohtani, E., & Maeda, M. (2001). Density of basaltic melt at high pressure and stability of the melt at the base of the lower mantle. *Earth and Planetary Science Letters*, 193(1–2), 69–75. https://doi.org/10.1016/S0012-821X(01)00505-2
- Oren, C., & Nowack, R. L. (2017). Seismic body-wave interferometry using noise autocorrelations for crustal structure. *Geophysical Journal International*, 208(1), 321–332. https://doi.org/10.1093/gji/ggw394
- Pham, T. S., & Tkalčić, H. (2018). Antarctic Ice properties revealed from teleseismic P wave coda autocorrelation. Journal of Geophysical Research: Solid Earth, 123(9), 7896–7912. https://doi.org/10.1029/2018JB016115
- Pham, T. S., & Tkalčić, H. (2021). Constraining floating ice Shelf structures by spectral response of teleseismic P-wave coda: Ross ice shelf, Antarctica. *Journal of Geophysical Research: Solid Earth*, 126(4), e2020JB021082. https://doi.org/10.1029/2020JB021082
- Qin, L., Ben-Zion, Y., Bonilla, L. F., & Steidl, J. H. (2020). Imaging and monitoring temporal changes of shallow seismic velocities at the Garner Valley near Anza, California, following the M7.2 2010 El Mayor-Cucapah earthquake. *Journal of Geophysical Research: Solid Earth*, 125(1), e2019JB018070. https://doi.org/10.1029/2019JB018070
- Qin, L., Qiu, H., Nakata, N., Deng, S., Levander, A., & Ben-Zion, Y. (2023). Variable daily autocorrelation functions of high-frequency seismic data on Mars. Seismological Research Letters, 94(2A), 746–758. https://doi.org/10.1785/0220220196
- Rivoldini, A., Van Hoolst, T., Verhoeven, O., Mocquet, A., & Dehant, V. (2011). Geodesy constraints on the interior structure and composition of Mars. *Icarus*, 213(2), 451–472. https://doi.org/10.1016/j.icarus.2011.03.024

- Samuel, H., Drilleau, M., Rivoldini, A., Xu, Z., Huang, Q., Garcia, R. F., et al. (2023). Geophysical evidence for an enriched molten silicate layer above Mars's core. *Nature*, 622(7984), 712–717. https://doi.org/10.1038/s41586-023-06601-8
- Schimmel, M., & Paulssen, H. (1997). Noise reduction and detection of weak, coherent signals through phase-weighted stacks. *Geophysical Journal International*, 130(2), 497–505. https://doi.org/10.1111/j.1365-246X.1997.tb05664.x
- Schimmel, M., Stutzmann, E., Lognonné, P., Compaire, N., Davis, P., Drilleau, M., et al. (2021). Seismic noise autocorrelations on Mars. *Earth and Space Science*, 8(6), 1–22. https://doi.org/10.1029/2021EA001755
- Scholz, J. R., Widmer-Schnidrig, R., Davis, P., Lognonné, P., Pinot, B., Garcia, R. F., et al. (2020). Detection, analysis, and removal of glitches from InSight's seismic data from Mars [Software]. Earth and Space Science, 7(11), e2020EA001317. https://doi.org/10.1029/2020EA001317
- She, Y., Yao, H., Yang, H., Wang, J., & Feng, J. (2022). Constraining the depth extent of low-velocity zone along the Chenghai Fault by dense array ambient noise interferometry and horizontal-to-vertical spectral ratio. *Tectonophysics*, 827, 229265. https://doi.org/10.1016/j. tecto.2022.229265
- Spohn, T., Grott, M., Smrekar, S. E., Knollenberg, J., Hudson, T. L., Krause, C., et al. (2018). The Heat Flow and Physical properties package (HP3) for the InSight mission [Dataset]. Space science reviews. https://doi.org/10.1007/s11214-018-0531-4
- Stähler, S. C., Khan, A., Bruce Banerdt, W., Lognonné, P., Giardini, D., Ceylan, S., et al. (2021). Seismic detection of the Martian core [Dataset]. Science, 373(6553), 443–448. https://doi.org/10.1126/science.abi7730
- Sun, W., & Tkalčić, H. (2022). Repetitive marsquakes in Martian upper mantle. *Nature Communications*, 13(1), 1695. https://doi.org/10.1038/ s41467-022-29329-x
- Thomson, W. T. (1950). Transmission of elastic waves through a stratified solid medium. Journal of Applied Physics, 21(2), 89–93. https://doi.org/10.1063/1.1699629
- Tibuleac, I. M., & von Seggern, D. (2012). Crust-mantle boundary reflectors in Nevada from ambient seismic noise autocorrelations. *Geophysical Journal International*, 189(1), 493–500. https://doi.org/10.1111/j.1365-246X.2011.05336.x
- Wen, L., & Helmberger, D. V. (1998). Ultra-low velocity zones near the core-mantle boundary from broadband PKP precursors. Science, 279(5357), 1701–1703. https://doi.org/10.1126/science.279.5357.1701
- Williams, Q., & Garnero, E. J. (1996). Seismic evidence for partial melt at the base of earth's mantle. Science, 273(5281), 1528–1530. https:// doi.org/10.1126/science.273.5281.1528
- Wookey, J., & Helffrich, G. (2008). Inner-core shear-wave anisotropy and texture from an observation of PKJKP waves. *Nature*, 454(7206), 873–876. https://doi.org/10.1038/nature07131
- Xu, Z., Broquet, A., Fuji, N., Kawamura, T., Lognonné, P., Montagner, J.-P., et al. (2023). Investigation of Martian regional crustal structure near the dichotomy using S1222a surface-wave group velocities. *Geophysical Research Letters*, 50(8), e2023GL103136. https://doi. org/10.1029/2023GL103136
- Yao, J., & Wen, L. (2014). Seismic structure and ultra-low velocity zones at the base of the Earth's mantle beneath Southeast Asia. *Physics of the Earth and Planetary Interiors*, 233, 103–111. https://doi.org/10.1016/j.pepi.2014.05.009
- Yoder, C. F., Konopliv, A. S., Yuan, D. N., Standish, E. M., & Folkner, W. M. (2003). Fluid core size of Mars from detection of the solar tide. Science, 300(5617), 299–303. https://doi.org/10.1126/science.1079645
- Zhang, Y., Yoshino, T., & Osako, M. (2023). Effect of iron content on thermal conductivity of ferropericlase: Implications for planetary mantle dynamics. Geophysical Research Letters, 50(7), e2022GL101769. https://doi.org/10.1029/2022GL101769
- Zhou, J., & Zhang, W. (2021). Extracting reliable P-wave reflections from teleseismic P wave coda autocorrelation. Journal of Geophysical Research: Solid Earth, 126(11), e2021JB022064. https://doi.org/10.1029/2021JB022064

References From the Supporting Information

- Huang, Q., Schmerr, N. C., King, S. D., Kim, D., Rivoldini, A., Plesa, A. C., et al. (2022). Seismic detection of a deep mantle discontinuity within Mars by InSight. Proceedings of the National Academy of Sciences of the United States of America, 119(42). https://doi.org/10.1073/ pnas.2204474119
- Verhoeven, O., Rivoldini, A., Vacher, P., Mocquet, A., Choblet, G., Menvielle, M., et al. (2005). Interior structure of terrestrial planets: Modeling Mars' mantle and its electromagnetic, geodetic, and seismic properties. *Journal of Geophysical Research: Planets*, 110(E4), E04009. https:// doi.org/10.1029/2004JE002271
- Yoshizaki, T., & McDonough, W. F. (2020). The composition of Mars. Geochimica et Cosmochimica Acta, 273, 137–162. https://doi.org/10.1016/j.gca.2020.01.011